Answering Reviewers:

**Reviewer #1:**

Thank you very much for your careful review and constructive suggestions with regard to our manuscript. Meanwhile, I am also very grateful to you for the specific modification strategy. I have made comprehensive and detailed changes according to your suggestions.

1. The problem of over-broad description of research significance.
   We have adjusted the significance of the study to a more specific description for the study population and purpose.

2. Questions on Form 5.
   Considering that the content shown in Table 5 is the comparison of different dimensions of psychological needs of parents of children with different characteristics, there are no redundant indicators and data, so data cannot be deleted. Splitting the table into two tables would not be complete. Therefore, we reorient the table text to portrait.

**Reviewer #2:**

Firstly, thank you very much for your careful review and constructive suggestions with regard to our manuscript. We regret there were problems with the English. The paper has been carefully revised by a professional language editing service to improve the grammar and readability.

**Reviewer #3:**

Thank you very much for your comments and suggestions. According to your comments and requirements, we have revised the original manuscript.

We have revised the manuscript to retain the modification history, and made a more specific description of the status of the children, that is, "stay in ICU within 72 hours after operation". The survey method has also been specifically corrected to "online questionnaire survey"; The description in the second sentence of the background section of the abstract has been revised to address the psychological needs of parents of children with complex congenital heart disease; As for the problems in table 2, 4 and 5, we have deleted and adjusted the content of the table so that we can understand the content of the table more clearly. There are too many references in the discussion about the interpretation results. We have made corrections and made corresponding explanations and put forward appropriate suggestions for the results of this study.